Author's response to reviews

Title: Traumatic Pulmonary Pseudocysts: Two Case Reports

Authors:

Bulent Kocer (drbkocer@gmail.com)
Gultekin Gulbahar (mdgultekin@gmail.com)
Nesimi Gunal (nesimigunal@yahoo.com.tr)
Koray Dural (koraydural@yahoo.com)
Unal Sakinci (unalsakinci@hotmail.com)

Version: 3 Date: 3 July 2007

Author's response to reviews:

Dear Editor,

Our manuscript revised in light of the reviewers' comments as belows.

REVISION DUE TO REVIEWER'S COMMENT

Reviewer 1:
1) A paragraph which about cyst forming diseases of lung is added to discussion section.
2) Typographical and grammatical errors were corrected.
3) The document is checked for any spelling or typing mistakes and terminological errors were corrected.
4) The manuscript improved and edited by native English speaker.

Reviewer 2:
1) The document is checked for any spelling or typing mistakes and terminological errors were corrected.
2) The literature reviewed extensively and a series of 14 cases is mentioned in the text.
3) More information about the cases discussed in the Case Reports section.
4) A possible table comparing different studies and case reports is added to Discussion section.